Author's response to reviews

Title: Efficacy of palliative radiotherapy for gastric bleeding in patients with unresectable advanced gastric cancer: a retrospective cohort study

Authors:

Chihiro Kondoh (hirochi1981@gmail.com)
Kohei Shitara (kouheis0824@yahoo.co.jp)
Motoo Nomura (mnomura@aichi-cc.jp)
Daisuke Takahari (dtakahari@aichi-cc.jp)
Takashi Ura (tura@aichi-cc.jp)
Hiroyuki Tachibana (tchbn@aichi-cc.jp)
Natsuo Tomita (ntomita@aichi-cc.jp)
Takeshi Kodaira (109103@aichi-cc.jp)
Kei Muro (kmuro@aichi-cc.jp)

Version: 2 Date: 30 December 2014

Author's response to reviews: see over
December 30th, 2014

Dear Editor

Please find our revised manuscript entitled “Efficacy of palliative radiotherapy for gastric bleeding in patients with unresectable advanced gastric cancer”, including two sets of tables and figures for the possible publication in the *BMC Palliative Care*. According to your advice, the manuscript was revised as follows: 1. Lines and pages were numbered and page breaks were removed, 2. Conclusions were separated from a section of discussion, 3. Authors’ contributions were inserted between the sections of competed interests and acknowledgement.

Thank you very much for your attention to this manuscript. We look forward to hearing from you in the near future.

Sincerely,

Chihiro Kondoh, MD
Corresponding author
Department of Medical Oncology
Japanese Red Cross Nagoya Daiichi Hospital
3-35 Michishita-Cho, Nakamura-Ku, Nagoya 453-8511, Aichi, Japan
Tel : 81-52-481-5111
FAX : 81-52-485-1187
e-mail: hirochi1981@gmail.com